Author's response to reviews

Title: Oxytocin and cholecystokinin secretion in women with colectomy

Authors:

Bodil Ohlsson (bodil.ohlsson@kir.lu.se)
Jens F Rehfeld (JensFRHO3561Rehfeld@rh.dk)
Mary L Forsling (bodil.ohlsson@kir.lu.se)

Version: 4 Date: 23 September 2004

Author's response to reviews:

Dear Editor,

1. We have now removed the short title on the first side.
2. A special section competing interests are included before acknowledgements.
3. The order of Tables has been moved to the last side.
4. We are very sorry, I have consulted 2 experts but noone is able to help us with the figures. Fig 1 was cropped by the Editorial support. We have tried to do Fig 2 in the same way. The text to the left is fixed from our computer program when we draw the figure, it seems not possible to move it. We resubmitt these fig. If not appropiate, Can We send You the original figures and You cropp them?

Sincerely Bodil Ohlsson